Author's response to reviews

Title: Shared Principles of Ethics for Infant and Young Child Nutrition in the Developing World

Authors:

Jerome A Singh (singhj9@ukzn.ac.za)
Abdallah S Daar (singhj9@ukzn.ac.za)
Peter A Singer (singhj9@ukzn.ac.za)

Version: 5 Date: 4 June 2010

Author's response to reviews:

Dear Editorial Team,

I have made the following amendments to the manuscripts, in line with your recommendations (my response in CAPS):

Reference 16: please provide a functioning link to website
DONE

References: Please make sure references are cited in numerical order. We notice references 10 and 11 are not cited within the main text. Please make sure references continue to be cited in numerical order once these references are cited.

THESE REFERENCES WERE ACCIDENTALLY OMITTED IN THE PREVIOUS REVISION. THEY HAVE BEEN REINSERTED. THIS DOES NOT CHANGE THE EXISTING REFERENCE SEQUENCE OR NUMBERING.

Additional files: all additional files must be mentioned in the text in numerical order
DONE

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Underlining: Please note that underlining is not supported in the final full text; you may wish to change it to bold.
DONE. UNDERLINED TEXT HAS BEEN CHANGED TO ITALICS

Email address: please include email addresses for all the authors on the title page, using the following format:
AB: abcd@institution.ac.uk
EF: efgh@generic.co.uk
IJ: IJKL@corporation.com
DONE

Thank you.
Jerome Singh
Corresponding Author